Author’s response to reviews

Title: Renal abnormalities among HIV-infected, antiretroviral naive children, Harare, Zimbabwe: a cross-sectional study

Authors:

Vongai Dondo Dr (vdondo@yahoo.com)
Hilda A Mujuru Dr (hmujuru@mweb.co.zw)
Kusum J Nathoo Prof (knathoo@mweb.co.zw)
Maxwell Chirehwa Mr (mchirehwa@yahoo.com)
Zivanai Mufandaedza Mr (zivanaimf@yahoo.co.uk)

Version: 2 Date: 12 June 2012

Author’s response to reviews:

Inserted page numbers

Under methods, page 5, Altered sentence about consent so that it now reads ’All the children had written informed consent from their parents/guardian and in addition, children aged 7 or older provided written informed assent"